Author's response to reviews

Title: Return to work after a workplace-oriented intervention for patients on sick-leave for burnout - a prospective controlled study

Authors:

Björn Karlson (bjorn.karlson@med.lu.se)
Peter Jönsson (peter_2_jonsson@med.lu.se)
Birgitta Pålsson (birgitta.palsson@med.lu.se)
Gunnel Åbjörnsson (gunnel.abjornsson@med.lu.se)
Birgitta Malmberg (birgitta.malmberg@med.lu.se)
Britt Larsson (brila@inr.liu.se)
Kai Österberg (kai.osterberg@med.lu.se)

Version: 4 Date: 20 May 2010

Author's response to reviews:

Dear Editor

We have now made the suggested major and minor revisions suggested, and re-submit the final version of the proof.

Major revision
The trial registration number is now included also in the abstract in the submission system.

Minor revisions
The table titles now appear above the tables and are one sentence long. Visible vertical lines are removed from the tables.

Minor typographical and linguistic corrections or clarifications are made throughout the manuscript.